Author's response to reviews

Title: Perforated carcinoma of the caecum presenting as necrotising fasciitis of the abdominal wall, the key to early diagnosis and management.

Authors:

Conor D Marron (cdmarron@mac.com)
Gerrarde T McArdle (gerrymeardle@hotmail.com)
Milind Rao (rmilind@hotmail.com)
Stephen Sinclair (stephen.sinclair@ucht.n-i.nhs.uk)
John Moorehead (john.moorehead@ucht.n-i.nhs.uk)

Version: 2 Date: 18 September 2006

Author's response to reviews:

Dear Sirs,

The paper has had minor amendments made in response to the suggestions from Reviewer 1. More specifically, I have adjusted the english language within the paper and removed capital letters within sentences where appropriate, particularly in references to necrotising fasciitis.

I look forward to hearing the outcome of the review and subsequent publication of this article,

Yours Sincerely,

Conor Marron